Author's response to reviews

Title: Leadership, staffing and quality of care in nursing homes

Authors:

Anders Kvale Havig (akh@nova.no)
Anders Skogstad (anders.skogstad@psysp.uib.no)
Lars Erik Kjekshus (l.e.kjekshus@medisin.uio.no)
Tor Inge Romøren (tromoren@online.no)

Version: 3 Date: 15 November 2011

Author's response to reviews:

1) The words 'title page' are removed from the title page
2) The title is removed from page 4
3) All references are cited in numerical order
4) Page 19: Two more references are included: 66 and 67
5) Page 21: One sentence is included: Furthermore, the literature review by Spilsbury et al. [13] concluded that: “…research has produced inconsistent and contradictory results about the link between nurse staffing and quality in nursing homes.” (p. 748).